Author's response to reviews

Title: Small RNA interference-mediated gene silencing of heparanase abolishes the invasion, metastasis and angiogenesis of gastric cancer cells

Authors:

Liduan Zheng (ld_zheng@hotmail.com)
Guosong Jiang (jiangguosong@yahoo.com.cn)
Hong Mei (meihong@126.com)
Jiarui Pu (pujiarui@yahoo.com)
Jihua Dong (jihuadong@126.com)
Xiaohua Hou (houxiaohua@yahoo.com)
Qiangsong Tong (qs_tong@hotmail.com)

Version: 3 Date: 4 February 2010

Author's response to reviews: see over
We thank the editors for their comments, excellent and inspiring suggestions regarding our manuscript. The responses to the concerns are embedded in each of the editors’ comments and concerns. We made changes accordingly and we hope that we have answered all the questions and the concerns from the editors.

**Major revisions:**

1. **Affiliations:** Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution

   **Response:** In this revised manuscript, we have provided the full affiliation for all authors as required.

2. **Figure 4:** We suggest you reduce the volume of text presented in the figure legends, as this will compete with the image for space in the final publication. Ideally figure legends should be approximately 250 words.

   **Response:** In this revised manuscript, we have reduced the text in figure legend of Figure 4 to 249 words as required.

**Minor revisions:**

1. **Running title:** please remove this

   **Response:** In this revised manuscript, we have removed the running title as required.

2. **Funding:** Please move the statement of funding to the Acknowledgements section and remove it from the title page

   **Response:** In this revised manuscript, we have moved the statement of funding to the Acknowledgements section as required.

3. **Email address:** please include email addresses for all the authors on the title page, using the following format: AB: abcd@institution.ac.uk  EF: efgh@generic.co.uk  IJ: IJKL@corporation.com

   **Response:** In this revised manuscript, we have provided the email addresses for all authors on the title page as required.

4. **Please present all text in black**

   **Response:** In this revised manuscript, we have presented all text in black as required.

5. **Key words: please remove these**

   **Response:** In this revised manuscript, we have removed the key words as required.
6. Introduction: please rename this section Background

Response: In this revised manuscript, we have renamed the “Introduction” as “Background” as required.

7. Materials and Methods: please rename this section Methods

Response: In this revised manuscript, we have renamed the “Materials and Methods” as “Methods” as required.

8. Please be consistent when citing figures, use either fig. or figure in both the main text and the figure legends section

Response: In this revised manuscript, we have cited the figures consistently in both the main text and the figure legends as required.

9. Figures: The image file should not include the title (e.g. Figure 1... etc.) or figure number. The legend and title should be part of the manuscript file after the reference list. The figures are numbered automatically in the order in which they are uploaded.

Response: In this revised manuscript, we have edited the figures as required.

10. Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

Response: In this revised manuscript, we have edited the figures as required.

11. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: In this revised manuscript, we have checked the typographical errors of manuscript and made changes accordingly as required.